Author’s response to reviews

Title: The feasibility and acceptability of collaborative learning in improving health worker performance on adolescent health: findings from implementation research in Moldova.

Authors:

Galina Lesco (galina.lesco@gmail.com)
Frances Squires (squiresf@gmail.com)
Viorel Babii (viorel.babii@gmail.com)
Nadejda Bordian (info@neovita.md)
Olga Cernetchii (nadejda007@mail.ru)
Adriane Martin Hilber (adriane.martinhilber@swisstph.ch)
Venkatraman Chandra-Mouli (chandramouliv@who.int)

Version: 1 Date: 11 May 2018

Author’s response to reviews:

All the requested revisions have been made.

Please note that specific ethics approval was not required for this paper since it arose from monitoring of a nationwide initiative approved by the Government of Moldova, rather than a research project. A letter confirming this can be provided if required.

Many thanks.